Author’s response to reviews

Title: Validity and reliability of measures to assess constructs from the Inner Setting domain of the Consolidated Framework for Implementation Research in a pediatric clinic network implementing HPV programs

Authors:

Timothy Walker (timothy.j.walker@uth.tmc.edu)
Serena Rodriguez (Serena.Rodriguez@UTSouthwestern.edu)
Sally Vernon (Sally.W.Vernon@uth.tmc.edu)
Lara Savas (lara.staub@uth.tmc.edu)
Erica Frost (Erica.L.Frost@uth.tmc.edu)
Maria Fernandez (Maria.E.Fernandez@uth.tmc.edu)

Version: 3 Date: 20 Feb 2019

Author’s response to reviews:

Thank you for continuing to consider our manuscript. Please see our responses below.

1. Ethics Approval and Consent

Please provide further clarification on consent to participate under 'Ethics Approval and Consent to Participate'.

Response: Consent was obtained for all study participants. We just received a waiver of documentation for consent. We added the following information to make this clearer: “We provided a letter of information about the study and by progressing through the letter to the survey and completing the questions, participants gave consent. This was done to maintain confidentiality of respondents so they felt comfortable answering the questions in an honest manner. This approach was approved by the UTHealth Institutional Review Board.”
2. Funding

Currently, you have stated "The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health."

Please add a sentence to state that the funding body/bodies had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We added a sentence stating the funding bodies have no role in the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: We have submitted a clean version of the manuscript.